RESEARCH





Public involvement in an aggregate and individual participant data meta-analysis of mindfulness-based programmes for mental health promotion

Claire Friedrich^{1,2}, Toni Fairbairn³, Graham Denton³, Mary Geddes³, Darren Thomas-Carr³, Peter B. Jones^{1,4} and Julieta Galante^{1,5*}

Abstract

Background Involving the public in evidence synthesis research is challenging due to the highly analytic nature of the projects, so it is important that involvement processes are documented, reflected upon, and shared to devise best practices. There is a literature gap on the involvement of the public in individual participant data meta-analyses, particularly in public health projects. We aimed to document and reflect on our collective experiences of involving and being involved as public stakeholders at all stages of a systematic review and individual participant data meta-analysis project.

Methods We formed a stakeholder group made of four members of the public at the beginning of our evidence synthesis project comprising a systematic review, an aggregate data meta-analysis, and an individual participant data meta-analysis of mindfulness-based programmes for mental health promotion in non-clinical adults. Following each group meeting, members and participating researchers completed written reflections; one group member collected and collated these. At the end of the project, a reflective writing workshop was held before all members completed their final reflections. Everyone completed an adapted, open-ended questionnaire which asked about what did and did not work well, the overall experience, what could be improved, and the felt impact the stakeholder group had on the research.

Results Overall, the stakeholders and researchers reported a positive experience of working together. Positives from the stakeholders' point of view included learning new skills, experiencing research, and making new friends. For the researchers, stakeholders helped them focus on what matters to the public and were reinvigorating research partners. The challenges stakeholders experienced included having long gaps between meetings and feeling overwhelmed. The researchers found it challenging to strike the balance between asking stakeholders to be involved and for them to learn research-related skills without overburdening them and making sure that the learning was engaging. When looking back at their experience, stakeholders described seeing their impact on the project in hindsight but that this was not felt while the project was being carried out.

Conclusion Successfully involving the public in complex evidence synthesis projects is possible and valuable from the points of view of the researchers and the stakeholders. However, it requires a significant time, skill,

^{*}Correspondence: Julieta Galante mjg231@cam.ac.uk Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.gr/licenses/by/4.0/.

Page 2 of 11

and resource investment that needs to be factored in from project inception. Further guidance and stakeholder training materials would be helpful. Specific suggestions are provided.

Keywords Patient and public involvement, Stakeholder involvement, Systematic review, Individual participant data meta-analysis, Mindfulness

Background

Patient and public involvement (PPI) in health research describes involving individuals with user knowledge and lived experience of the research topic who do not primarily carry out the research [1]. Non-profit organisation INVOLVE defines public involvement as research that is carried out by or with members of the public rather than about or for them [2]. PPI group members (from now on 'stakeholders') can include professionals, patients and caregivers, or the public, although the latter is less common. Stakeholders are invited to bring lived experiences to the research with the goal of improving the relevancy of research aims and quality [3]. Benefits of stakeholder involvement include prioritising research questions, selecting more relevant outcomes, creating innovative solutions and new ideas, and ensuring equality of public access to policy and decision-making [3]. It is becoming more common practice to involve stakeholders in health research in the UK, and stakeholder involvement is now fundamental to government-funded health research [4]. Stakeholders can be invited to join research projects of any methodologies, including intervention trials, qualitative research, or evidence syntheses such as systematic reviews (SRs) [5]. Stakeholders will engage in a wide range of research activities in keeping with the project aims and methodologies.

When involved in intervention studies, they may support guiding the intervention design, recruiting participants, or selecting outcomes [5]. However, engaging the public in evidence synthesis projects is more challenging because there are no interventions to design and no participants to recruit; instead, most of the processes tend to be quite technical, requiring specific research skills. Still, stakeholders engaged in SR projects have contributed by finalising research questions and protocols, selecting outcomes, making sense of findings, and supporting dissemination tasks [6, 7]. The benefits of stakeholder involvement in SRs include increasing the relevancy of the research, providing a real-world context, and enhancing end-user uptake of the interventions or treatments reviewed [8]. Involving stakeholders can improve the actual and perceived usefulness of SRs and is likely to reduce research waste [9].

Researchers are incorporating stakeholder groups within their projects at increasing rates [10], leading to clearer recommendations around involving stakeholder members and for reporting on and assessing the impact of their involvement [10]. However, evidence and guidance are limited specifically within the context of evidence synthesis projects and when involving public, less so professional or patient, stakeholders [1]. Additionally, the level and clarity of reporting are inconsistent with reviews not reporting stakeholder characteristics or their involvement activities, and none measured the impact of involving stakeholders [1].

There are developments around stakeholder impact guidance and frameworks, such as the Public Involvement Impact Assessment Framework [11] and the Guidance for Reporting Involvement of Patients and the Public (GRIPP2) checklist [12, 13], however a lack of information on actual stakeholder involvement throughout the SR process remains [9]. In response to this gap, the ACTIVE framework (Authors and Consumers Together Impacting on Evidence) has been developed and can be accessed as online training via Cochrane [14]. It covers the essentials of good practice such as defining stakeholder roles and responsibilities and forming good relationships between researchers and stakeholders through respect, trust, confidentiality, and clear communication [9]. These recommendations are in keeping with the NIHR INVOLVE values and principles framework [2] which Greenhalgh et al. [15] use as an example of a partnership-focused framework due to its standards set around inclusive opportunities and working together in respectful, supportive, and transparent ways.

Greenhalgh et al. [15] synthesised 65 stakeholder involvement frameworks and report large heterogeneity and limited transferability across these frameworks indicating that a 'one-size-fits-all' framework may not be suitable for guiding researchers in their design and implementation of stakeholder involvement. Instead, they suggest making evidence-based resources available for researchers to co-design their own framework with project stakeholders [15], using a co-design approach could certainly make the framework more applicable to individual project and allow for more input from stakeholders. If using an approach without established frameworks, researchers should have access to reports on stakeholder and researcher experiences to support their decision-making and co-design of their own framework.

Due to the limited evidence and reporting of stakeholder experiences in evidence synthesis work, it is recommended that researchers carrying out evidence synthesis projects collect data on the stakeholder experience from the stakeholders themselves [9]. Not many reviews have described the experience of researchers, and even less so of the stakeholders themselves, when involving stakeholders in evidence synthesis projects. In one qualitative meta-synthesis project, stakeholders received training, helped develop a coding frame and attended a focus group to support the researchers in interpreting the data [7]. Stakeholders provided mixed feedback on their ability to understand research and medical terminology throughout the project and in the training material; they wanted face-to-face training and feedback from the researchers to know how they were doing [7]. Stakeholders gave positive feedback and happily be involved in future projects, although questioned the benefit and extent of their contribution [7], which is similar to the feedback collected by Vale et al. [16].

In an individual participant data meta-analysis (IPD-MA) patient stakeholders, named 'Patient Research Partners' were involved in recruiting professional stakeholders, providing feedback on documents, contributing to project newsletters, and providing input into the lay summary for the Cochrane review [16]. The group in general had a positive experience and all would do it again; however, as with Bayliss et al. [7], some questioned the usefulness of their input in the context of a SR as the outcomes are pre-set to what trials had already collected [16].

It is necessary to capture feedback from stakeholders, because reviews which do not will miss out on a richer level of information and are not able to provide guidance for future researchers to help them with planning their training material and method of stakeholder involvement. This paper examines the experience of public stakeholders within a 4-year long, two-stage SR project looking at mindfulness-based interventions for mental health promotion in a non-clinical sample. The findings of each stage are published elsewhere [17, 18]. The first stage involved completing a SR and aggregate data metaanalysis (AD-MA) of 136 randomised controlled trials (RCTs) of mindfulness-based interventions for mental health promotion in community adults [17]. Then we completed an IPD-MA with the data of 13 RCTs with the same intervention and population types [18, 19]. Public stakeholders were involved at all stages, including in coauthoring this paper.

Following the involvement of a stakeholder group in their SR and IPD-MA project, Vale et al. [16] encourage future researchers to report on the involvement and experiences of stakeholders in their reviews. While Bayliss et al. [7] recommend further research to monitor and evaluate stakeholder involvement, both of which we intend to do. We aim to address the current lack of evidence and guidance of public stakeholder involvement in evidence synthesis projects by documenting and analysing reflections of our collective experiences of involving and being involved as stakeholder members. Researchers and stakeholders co-author this paper to fully understand their experience and to help build evidence around what might be helpful or unhelpful in involving stakeholders in evidence synthesis projects. As the stakeholder group joined a project carrying out specifically an AD-MA and IPD-MA, this reflective work is carried out within the context of these particular types of MAs. To our knowledge, this is the first paper to examine the experience of non-clinical stakeholders' involvement in an IPD-MA project.

Methods

Forming the stakeholder group

Members of the public were invited to join researchers within the Department of Psychiatry at the University of Cambridge. The PI (JG) looked to recruit people with diverse experiences and interests in mental health promotion and mindfulness. Potential stakeholder members were recruited through personal networks and social media advertisements in Spring 2018.

As the SRs conducted examined non-clinical adult samples, the stakeholders were not recruited for having any particular clinical characteristics. However, there was a requirement to achieve diversity in terms of gender, age (aside from having to be adults over 18 years old), and experience with mindfulness practices. Therefore, stakeholders were carefully chosen. In addition, due to the requirement to hold in-person meetings in university buildings in Cambridge, recruited stakeholders had to be local to the Cambridgeshire area. They were given information by the PI about the study and expectations, such as anticipated time commitment, and had the opportunity to discuss the role in detail before committing.

The resulting stakeholder group was made up of two men and two women, ages ranged from 40 to 60s, with varied professional and educational backgrounds. One member had previously attended a Mindfulness-Based Stress Reduction (MBSR) course, one member had a long-term mindfulness meditation practice, while another member had no mindfulness experience (yet experienced in other contemplative practices), and finally, one member had had a dissatisfying initial engagement with mindfulness meditation but was very interested in mental health promotion.

Professional stakeholders were additionally recruited to form an advisory group with the help of public stakeholders. Professional stakeholders ranged in professional background and in terms of personal experiences and opinions of mindfulness. They were kept up to date and their input was sought throughout the study; however, this paper focuses on the experience of public stakeholders so unless specified otherwise, the use of the term stakeholders remains in reference to our public stakeholder group.

Running the stakeholder group

Stakeholders were involved from the project's outset. Regular meetings were organised by the PI throughout: two workshops and ten group meetings took place in total from June 2018 to July 2022. In-person meetings lasted between 2 and 4 h. From March 2020, meetings shifted online in response to Covid-19 (with shorter durations) and were once again, conducted in-person from October 2021 onwards. Members were able to attend most meetings and workshops and were reimbursed for their time and transport immediately after attending.

As members were new to health research, about half of the time in initial meetings was used for induction by the PI, covering research background and project aims, timelines, and analyses and explaining the stakeholder role. The training also covered research methodology, specifically RCTs and SRs with MAs. The PI put together an induction document explaining research background, and a glossary. She also encouraged stakeholders to complete Cochrane Collaboration training on SRs for stakeholders [20]. Later, meetings were used to provide study updates and results, and stakeholders were encouraged to provide feedback. Two workshops relating to effective communication (suggested by the PI) and reflective writing (collectively decided) were organised with external workshop leaders in which researchers and stakeholders were trained together. Members of the group also acted as research partners throughout and took part in different SR and dissemination tasks.

Stakeholders as research partners

Stakeholders were involved in the SR process after receiving training from the PI and a research assistant. Two members were involved in the title and abstract screening using Covidence [21]. Then, following further training, an additional member joined for the data extraction stage, again using Covidence.

After publication of the SR and AD-MA [17], all members stayed on as stakeholders while the IPD-MA was carried out and published [18]. Stakeholders also participated in co-producing a project film, where the review process was summarised, and findings were disseminated. They were involved in the planning and scriptwriting, as well as the acting and voice-over. Their final stakeholder task was to provide reflections on their experiences as stakeholders and to complete this reflective paper together with the researchers. One stakeholder (TF) oversaw collecting the post-meeting reflections and final evaluation questionnaire and analysed these for this paper.

Stakeholder and researcher reflections

Throughout the project, stakeholders and researchers completed short, personal reflections following each meeting and submitted these to the stakeholder in charge per email. These reflections were to outline their thoughts and experiences in their project involvement and to later write this reflective paper on their experiences. Additionally, nearing the end of the project, a reflective writing workshop was organised as stakeholders and researchers expressed a need for it. Following the workshop, stakeholders and researchers completed an adapted version of the Stakeholders Evaluation Questionnaire designed and shared with us by Vale et al. [16]. The questionnaire was adapted with the help of one stakeholder (TF), and guestions covered a range of experiences, asking about their motivation for involvement as a stakeholder, the quality of information and training they received, and their thoughts on their level of involvement.

Additionally, they were asked about the stakeholder group itself, their thoughts on the group size and diversity, the length and frequency of meetings, and their expectations at the project start and if these were met. Reflective questions asked about what the best and worst aspects of being involved in the project were, what their felt impact on the research was, and whether they would do it again or recommend getting involved in SR and MA projects to other people. Suggestions for improving future stakeholder recruitment, training, and involvement were also sought. Members were repeatedly reminded to be honest in their responses and that the researchers would not be upset by negative feedback. A summary of stakeholders' and researchers' experience of stakeholder involvement in the project was established through combining the questionnaire responses and post-meeting feedback. One stakeholder (TF) and the researchers gathered individual questionnaire answers and then grouped together responses into emerging themes guided by the existing questionnaire items and stages of stakeholder involvement (e.g., experience at the start of being involved and reflections looking back over their involvement when the project was completed). The emergent themes and selected quotes to support these were finalised through wider group and PI feedback.

Results

All four stakeholders and the PI completed the reflections questionnaire in October 2022. Stakeholders, the PI, and researcher also completed reflections following group meetings and workshops throughout the course of the project. On average, group members completed four post-meeting reflections, and everyone completed the Stakeholder Evaluation Questionnaire [16].

Information provided and level of understanding among stakeholders

Stakeholders reported that, despite being unfamiliar with the research process and methodologies, the PI's explanations about key research concepts and the use of simple but interactive exercises made their involvement a positive experience. They felt that alongside the well-balanced explanations and background information provided by the researcher, frequent breaks and opportunities to ask questions made it more stimulating and enjoyable.

Three of the four group members initially voiced concerns that without a scientific background, they would feel out of their depth in understanding research concepts. However, after attending several meetings, all members reported that they had been made to feel at ease through clear and helpful presentations, which were enhanced by interactive role play and workshops:

Beforehand I thought that I might be out of my depth, being a non-academic, but this was not generally the case. I was struck by how a collaborative and consensual framework was established and that we were taken seriously, so that I felt a valued member of the group rather than a token requirement of the project. -Stakeholder

Two group members expressed feeling overwhelmed by the statistics described and believed that it would not be possible for them to understand the analyses despite the presentations and explanations provided. This feeling persisted throughout the 4-year collaboration. An additional member felt that the face-to-face group meetings were well thought out and that the PI's guidance helped them understand key research concepts:

The workshops contained some really quite challenging concepts, such as statistical analysis.... however, the explanations were very accessible, and I thought that the practicals were quite entertaining, as well as informative. -Stakeholder

The most challenging meeting of the project was, from most of the stakeholders' viewpoints, the one to discuss the research protocol. There was a lot of new information to read through, which was challenging to all apart from the one stakeholder who had a scientific background. However, the three members who partook in literature screening and data extraction reported that it was a helpful meeting as it helped familiarise them to the project and research process. The PI expressed concerns around training stakeholders to a level at which they could conceptually understand all the steps of the research and aimed to make the training engaging and didactic. However, over time, the PI's view was that the group internalised the project more, with members finding their feet more quickly and making more specific suggestions, showing a developing understanding of research terminology and concepts.

Feelings about stakeholder involvement

All members responded positively to the question regarding how they felt about their involvement as public stakeholders in the project. They described varying reasons for why they rated their experience as positive, including learning a lot about research and the motivation behind it:

I found helping with the research, albeit in a small way, brought the project to life and instilled a better understanding of the why's and wherefores of systematic reviews. I would have welcomed a bigger role if the opportunity arose. -Stakeholder

The PI described the involvement of stakeholders positively and viewed the group's impact as important for varying reasons, including in providing helpful feedback, keeping the research relevant to the public, and holding the researchers accountable throughout the long project. The project had been a solitary activity at times so the PI felt accompanied throughout by the stakeholders, as they remained fully involved at all stages and provided a sense of teamwork and support:

There were many great things about involving stakeholders - such as getting lay viewpoints and feedback, getting help with the research itself, meeting new people, learning how to train and engage the public, and having fun during meetings... Also, being accountable to them helped me structure and keep it relevant. I was pleasantly surprised at how refreshing and helpful it can be to share your project with enthusiastic members of the public. – Researcher

The PI expressed some regret concerning a lack of time to involve the stakeholders as much as she wanted and reported that the balance between doing the research and involving stakeholders, due to limited time, had been difficult.

What went well

All four group members stayed on through the course of the study and reported very positive experiences overall, reporting that they shared a new experience, learned new skills, and made new friends: Learning new skills, doing something completely different, de-mystifying academia, putting a human face on it, meeting new people from other walks of life..." -Stakeholder

I felt that my favourite thing about the project was being part of a cutting-edge study as to the benefits, or not, of mindfulness. I have never had the opportunity to take part in a stakeholder group and it opened up a whole new area for me with the added blessing of making new friends and providing stimulus regarding a debated subject. Unfortunately, as the experience draws to a close it does leave a void. Overall, it was a real privilege to have been given the opportunity to take part. I would love to be involved in future stakeholder groups. -Stakeholder

All four agreed that meeting new people of different ages and backgrounds was positive and stimulating and that the group blended well and became a friendly and cohesive group.

What did not go well

The group was aware that the research study, from its outset to conclusion, would take 4 years. However, there were drawbacks to the long breaks between meetings where members felt they had lost some interest in the study and were forgetting details. Two stakeholders suggested some more frequent group meetings:

There were long gaps, during which time I lost interest and forgot where we were at. I wonder if it would have been feasible to organise mini catchups in between where we were required to read through the notes from previous meetings beforehand? -Stakeholder

Another challenge reported by a member was the duration of the meetings as they found that 4-h meetings were difficult, particularly when covering more challenging topics such as statistics. Two members additionally described grappling with multiple documents and PowerPoint to be difficult at times.

The PI would have preferred to hold more regular and frequent meetings. The challenge was to balance the time to carry out the research with the time to meet with stakeholders. It was also deemed unnecessary to convene a meeting when there were limited things to present or discuss with the group. Another point the PI had to consider when planning and running the sessions was that group members had different abilities and understanding, so there was a risk of imbalance which could lead to disinterest by some if it was not as engaging, but frustration for others if too challenging: At times, the concepts were difficult to grasp so perhaps some became disengaged or struggled to follow the session for a bit, while others could follow me. I tried to accept this as normal; a diverse group means different people will understand different things better. -Researcher

Felt impact of public stakeholders on research

Stakeholders initially felt their efforts carried little impact throughout the research project; however, by the end of the project, they were able to see some evidence of their involvement on the study outcome:

When we reflected recently, I didn't feel I'd made very much impact on the research. However, reading the summary of the project, it does sound like the feedback from our group has helped guide decisions in how the research has proceeded. -Stakeholder

Other stakeholders reported feeling that the group's input and feedback may have prompted the lead PI to do some things differently. One group member described feeling unsure of the group's impact but hoped that their life experiences and views shared in meetings were able to represent the wider public. It was suggested by members that it would be useful to frequently remind stakeholders about how they are contributing to the project.

While she tried to transmit the usefulness of their contributions informally at each meeting, the PI wondered if one of the reasons that the stakeholders felt that they may not had contributed much to the project was because she did not have a formal and consistent process of giving them detailed feedback on how she had used their suggestions. This lack of formal process was down to the researchers' lack of time, and the need to prioritise other aspects of the stakeholder involvement with the time available.

Study outcome

The group had mixed views about the outcomes of the research. Some said they had no real expectations of the outcomes and were hopeful, but in one case, sceptical about its benefits. One participant was disappointed when the conclusions of the first phase of the study were shared:

I initially felt disappointed that the meta-analysis did not show a strong benefit from mindfulness, compared to other interventions, such as exercise. I was expecting a clearer benefit to be seen. However, I think the results resonate with my own experience, in that I find exercise to be more beneficial for my mental wellbeing than meditating. -Stakeholder **Would they do it again and recommend the role to others?** All four stakeholders said they would be happy to be public stakeholders again, and, with caveats, would recommend the role to others. While positively recommending the role, one stakeholder stated:

...in many ways meta-analysis is an end-game and it may be that individual research projects themselves would be more attractive to some. My advice would be not to be put off if, initially, stakeholders felt out of their depth. This is to be expected if medical research is a completely new experience. A good lead researcher helps allay that feeling by engaging well with the group, simplifying explanations and making meetings stimulating and fun. -Stakeholder

Another member reflected that, ideally, stakeholders should have a general interest in the research subject, and it was important to approach the project with an open mind on the findings—not feeling too invested in arriving at a particular conclusion. It was also important to consider the time pressures involved for stakeholders who have busy working lives.

Feelings about the group

The group felt the four-member number was adequate, although some pointed out that up to two more participants would, perhaps, have added more diversity and would have been useful in case one or more stakeholders had dropped out along the way.

Involvement of public stakeholders in future research

With the right amount of time and resources available, the researchers would involve members of the public again in future research. The PI highlights the importance of involving stakeholders as soon as possible, including when putting together the grant application and drafting research protocols. The stakeholder feedback on the research plans and protocol was perceived as being just as essential as their involvement throughout the study so researchers would benefit from early stakeholder involvement. When undertaking an IPD-MA, the PI advises other researchers to involve public stakeholders and to design participation in ways that provide an engaging and interesting experience:

It's essential to make meta-analysis training as engaging as possible, even if this takes more time and effort. I would also make sure to manage expectations and to remind stakeholders frequently about how they are contributing.

-Researcher

Discussion

Overall, the stakeholders and researchers reported a positive experience of working together throughout this two-stage complex evidence synthesis project. Numerous benefits and some challenges were reported throughout the feedback provided during the 4 years of involvement and within the final questionnaires. Stakeholders reported positively on their involvement, and all said they would participate in a similar project in the future. Positives from the stakeholders' points of view included learning new skills and making new friends and they generally enjoyed taking part in a research project. The researchers appreciated working with members of the public to help them focus on what matters to the public, found it helpful for stakeholders to carry out research tasks and enjoyed having meetings with an engaged group that brought fresh perspectives to the table.

The challenges stakeholders experienced included having long gaps between meetings, resulting in some waning interest and forgetting project details, and feeling overwhelmed at times with the level of information being described.

Similarly to the stakeholders, the PI would have preferred to have more frequent meetings, but time and funding were limited, so this was not possible. The PI additionally expressed difficulty with getting the teaching level right, particularly as stakeholder backgrounds ranged in experience of research. It was challenging at times to strike the balance between asking stakeholders to be involved and for them to learn research-related skills without overburdening them and making sure that the learning was palatable and engaging.

When looking back at their experience, stakeholders described seeing their impact on the project in hindsight but that this was not felt while the project was being carried out. The general feedback on how things could be improved in future evidence synthesis projects with stakeholders was to involve more stakeholder members, to hold more regular meetings, and to keep these shorter in duration when covering challenging topics such as statistics. It was also suggested that researchers could feed back to the group the influence stakeholders have on the research itself during the project so that they are aware of their impact and feel more useful in their contribution. And finally, despite the efforts of the PI to keep the learning as engaging and not overwhelming as possible, it was a challenge for some to keep up and to use some of the technology, such as Microsoft PowerPoint; hence, the importance of gauging the audience level prior to creating learning material and employing technology that all members are able to use.

The main challenge which came up throughout the course of the project and in the feedback questionnaires was that of overwhelming stakeholders with research terminology and concepts. The PI also found it challenging to have a group with mixed engagement and interest, but it seemed that after a while, stakeholders settled into their roles and became more comfortable with research concepts, although the feeling of uncertainty did persist for two group members. Stakeholders often report enjoying learning new skills [22]; however, researchers should remain vigilant of not burdening stakeholders and turning their experience negative. Generally, stakeholders will partake in projects on a very part-time basis so researchers should consider the challenges stakeholders may face when having to learn new concepts surrounding work, they may have limited interest and time for and to set realistic and achievable learning goals. It is important to understand if stakeholders are unhappy because they may disengage; an alternative could be that those who feel less comfortable with certain concepts are tasked with something they prefer. The group were not expected to apply statistical skills at any point, and rather were tasked with supporting the screening and data extraction which they did pick up more easily and reportedly enjoyed.

Numerous papers have described stakeholders struggling with research concepts and terminology and the stakeholders from Bayliss et al. [7] asked for something like a glossary, which Brütt et al. [5] provided for theirs. It could be that at the start, as Bayliss et al. [7] have done, the group starts with standardised training material and this is slowly developed as the group learns and adapts it to their level of knowledge and skill. An option could be to host a live glossary of terms that people can refer to and when new terms come up, researchers and stakeholders could add terms and definitions to these. However, stakeholders for this project did not feel it was necessary to have a live document. It is fairly evident that different groups have different preferences so researchers should use the initial meetings as an opportunity to ask stakeholder preferences and to co-produce or adapt learning material to their specific needs. This would also be a chance for researchers to understand the group's level of research knowledge and for stakeholders to communicate their needs.

In general, stakeholders involved in evidence syntheses have reported positive experiences and said they would all do it again [7, 16]. Members were reimbursed for their time, including their preparation for workshops or meetings; this is important as stakeholders should be paid for the experience and feedback provided [23]. This may perhaps also explain parts of the positive feedback received from stakeholders when they are reimbursed appropriately. This shows that factoring in costs for stakeholder activities is important in evidence synthesis project proposals so that stakeholders can be recruited and may be more likely to stay on throughout the entirety of the project.

The challenge associated with questioning the impact of their involvement as stakeholders has been repeatedly brought up by stakeholders of previous evidence synthesis projects [16, 24]. This feeling may be more present for stakeholders taking part in evidence synthesis projects, as opposed to intervention trials, as the outcomes are already pre-selected and reported in published trials which leaves limited room for input, particularly within quantitative evidence synthesis [24, 25]. A suggested solution to this is to involve stakeholders early on in projects, as early as the funding application or protocol registration, if resources allow, to not only benefit the stakeholder-perceived impact and subsequent experience, but also the true impact and subsequent review quality and relevancy. One of the most important impacts stakeholders had in our project was outcome selection, thanks to involving them from inception. Stakeholders in this project mentioned that in hindsight, they were able to perceive their impact but that during the project they questioned it. It seems like it would be beneficial to stakeholders to receive ongoing feedback from researchers on how they are shaping and influencing the project and to see the impact they have.

Additionally, as stakeholders seem to frequently question their effects on the evidence synthesis research project and outcome, and it is not often formally measured [26], it appears essential for researchers to have a measure of impact developed. A formal measure for stakeholder impact in evidence synthesis projects would be helpful for researchers, for funding bodies, and for stakeholders themselves [26]. Furthermore, there is a lack of collecting data and reporting on stakeholder experiences. The more researchers report on the actual experience of stakeholders, rather than reporting on their behalf, the more in-depth feedback we will have to help guide and improve the experience for future stakeholders.

The challenges reported by stakeholders and researchers in this paper pertaining to research tasks and logistics, such as payments and meeting attendance, are transferrable across the majority of evidence syntheses projects; however, researchers should be aware of unique challenges reflecting their SR PICO (Population, Intervention, Comparator, Outcome) they may encounter. Our stakeholder group was nonclinical and the SR looking at mental health promotion mindfulness interventions, however, researchers working with patient stakeholder groups (e.g. involving stakeholders with depression) should plan ahead for potential challenges catering to different support needs. When including interventions which may be slightly more technical in the terminology used in associated publications (e.g. examining underlying mechanisms of pharmaceutical treatments), researchers may need to develop more in-depth training for stakeholder members and dedicate more time to making the project information accessible.

Stakeholders made a clear difference in the research process by carrying out record screening and data extraction, which reduced the researcher workload. They additionally were involved in the co-production of a project film which has been shared publicly. Their involvement made the film content and language accessible, which may have increased the impact of the study output by reaching a wider audience.

Finally, their help in planning out the SR and MAs, including outcome, control group, and subgroup analysis selection, has potentially improved the generalisability and applicability of research findings due to being guided by those with experience of the intervention and for whom the findings would be relevant to. The feedback from the stakeholders also led to the PI seeking further clarifications from a sociologist to best strategise the ways in which the countries in which the interventions were carried out could be grouped for subgroup analysis; this led to a more appropriate cultural subgroup analysis.

There are currently limited guidelines for researchers to involve public and patient stakeholders in their evidence synthesis projects. As the number of evidence synthesis projects continues to grow [27] and research funding bodies emphasise the need for public and patient involvement in projects [28, 29] it is important to have these guidelines in place to ensure a positive experience for all involved parties. To develop guidelines, it is essential to have data on the impact and experiences from those involved, which is what we hope to contribute herein.

Moving forward, it is anticipated that future research will involve stakeholders at increasing rates in evidence synthesis projects and that researchers will endeavour to enhance involvement impact and experience through more in-depth assessments and reporting on stakeholder involvement. Measuring the impact of stakeholder involvement during a research project may become more established and evidence-based, as funders may become interested in measuring the impact in the longer term to justify costs of stakeholder involvement. To measure longer-term impact, it could be useful to compare the impact of projects which have been carried out with and without stakeholder involvement. As some of the justifications for stakeholder involvement include higher impact due to higher ecological validity and applicability (e.g. higher intervention uptake), it could be beneficial to examine the use of the evidence from projects with versus without stakeholder involvement.

It would also be useful for researchers and funders to see the influence that stakeholder involvement has on the wider general population understanding of and interest in science and healthcare research. Important to note for researchers considering stakeholder involvement in future evidence synthesis projects is the rapid rise of popularity of data-driven healthcare [30], use of artificial intelligence (AI) in evidence syntheses [31], and living systematic reviews (LSRs) [32] as these introduce unique considerations when planning and carrying out work with stakeholders. For example, involving AI and public stakeholders in screening activities simultaneously in a project may require more thorough training for stakeholders to guarantee screening accuracy so that the data can be used to train the AI model for screening. Finally, reporting on stakeholder involvement in SRs may become more mainstream which in turn may provide further evidence to develop quality guidelines for involving stakeholder members.

Limitations

We repeatedly asked group members to be as honest as possible in their feedback, but there remains the chance that responses deviate from actual experience to prevent giving negative feedback to the researchers. Unfortunately, we were unable to measure and report on the long-term impact of stakeholder involvement in an evidence synthesis project. Additionally, for approximately 18 months, in response to the Covid-19 pandemic regulations in the UK, we met online to carry out meetings and workshops. Effort was placed in maintaining the same dynamic as in-person training and meetings; however, as participants in Bayliss et al. [7] report, face-to-face training was seen as more beneficial and as a preferred method of meeting so this may have impacted our stakeholder experience.

Conclusion

To our knowledge, this is the first IPD-MA project with public stakeholder involvement. A limited number of IPD-MA projects have involved patient stakeholders and there has been some public stakeholder involvement in AD-MAs. Overall, involving a public stakeholder group in the SR process was a positive experience for the researchers and stakeholders, with minor caveats. However, these caveats, such as challenging training concepts or a sense of low impact by the stakeholders, could be better managed by the researchers if there were guidelines and established training material available. Successfully involving the public in complex evidence synthesis projects requires a significant time, skill, and resource

Abbreviations

Abbreviations	
PPI	Patient and public involvement
SR	Systematic review
MA	Meta-analysis
AD-MA	Aggregate data meta-analysis
IPD-MA	Individual participant data meta-analysis
RCT	Randomised controlled trial
PI	Principal investigator

- MBSR Mindfulness-based stress reduction
- Al Artificial intelligence
- LSR Living systematic review

Acknowledgements

Thank you to Lorna Jacobs, Dr Claire Vale, and Professor Jane Tierney for their PPI advice at different points in the project.

Authors' contributions

JG applied for research funding and is the guarantor. JG organised and led the stakeholder meetings; PBJ provided high-level advice. TF and JG adapted the reflective questionnaire. TF collected and TF and CF collated the reflections following meetings. CF, TF, and JG wrote the manuscript that was revised through discussion with all the authors. All authors read and approved the final manuscript.

Funding

This publication presents independent research funded by the National Institute for Health Research (NIHR). The views expressed are those of the authors and not necessarily those of the NHS, the NIHR, or the Department of Health and Social Care. All research from the Department of Psychiatry at the University of Cambridge is supported by the NIHR Cambridge Biomedical Research Centre (BRC-1215–20014) and the NIHR East of England Applied Research Centre. JG was funded by a NIHR Post-doctoral Fellowship for this research project (PDF-2017–10-018, salary and all project costs). CF's salary for this research project was funded by a Cambridgeshire and Peterborough NHS Foundation Trust grant awarded to JG. PBJ was supported by the Wellcome Trust (095844/Z/11/Z), the UK Medical Research Council (MR/N019067/1), and the NIHR ARC East of England.

Availability of data and materials

Data sharing is not applicable to this article as no datasets were generated or analysed during the current study.

Declarations

Competing interests

The authors declare that they have no competing interests.

Author details

¹Department of Psychiatry, University of Cambridge, Douglas House, 18B Trumpington Road, Cambridge CB2 8AH, UK. ²Department of Psychiatry, University of Oxford, Oxford, UK. ³Cambridge, UK. ⁴National Institute for Health Research (NIHR) Applied Research Collaboration East of England, Cambridge, UK. ⁵Contemplative Studies Centre, Melbourne School of Psychological Sciences, Faculty of Medicine, Dentistry, and Health Sciences, University of Melbourne, Australia.

Received: 29 October 2023 Accepted: 2 July 2024 Published online: 06 August 2024

References

 Pollock A, Campbell P, Struthers C, Synnot A, Nunn J, Hill S, et al. Stakeholder involvement in systematic reviews: a scoping review. Syst Rev. 2018;7(1):208.

- 2. What is public involvement in research? | INVOLVE [Internet]. 2012 [cited 2023 Oct 15]. Available from: https://www.invo.org.uk/find-out-more/ what-is-public-involvement-in-research-2/.
- Pizzo E, Doyle C, Matthews R, Barlow J. Patient and public involvement: how much do we spend and what are the benefits? Health Expect. 2015;18(6):1918–26.
- Hyde C, Dunn KM, Higginbottom A, Chew-Graham CA. Process and impact of patient involvement in a systematic review of shared decision making in primary care consultations. Health Expect. 2017;20(2):298–308.
- Brütt AL, Meister R, Bernges T, Moritz S, Härter M, Kriston L, et al. Patient involvement in a systematic review: development and pilot evaluation of a patient workshop. Z Für Evidenz Fortbild Qual Im Gesundheitswesen. 2017;127–128:56–61.
- Boote J, Baird W, Sutton A. Public involvement in the systematic review process in health and social care: a narrative review of case examples. Health Policy. 2011;102(2–3):105–16.
- Bayliss K, Starling B, Raza K, Johansson EC, Zabalan C, Moore S, et al. Patient involvement in a qualitative meta-synthesis: lessons learnt. Res Involv Engagem. 2016;2(1):18.
- Gierisch JM, Hughes JM, Williams JW, Gordon AM, Goldstein KM. Qualitative exploration of engaging patients as advisors in a program of evidence synthesis: cobuilding the science to enhance impact. Med Care. 2019;57(Suppl 3):S246–52.
- Pollock A, Campbell P, Struthers C, Synnot A, Nunn J, Hill S, et al. Development of the ACTIVE framework to describe stakeholder involvement in systematic reviews. J Health Serv Res Policy. 2019;24(4):245–55.
- Concannon TW, Grant S, Welch V, Petkovic J, Selby J, Crowe S, et al. Practical guidance for involving stakeholders in health research. J Gen Intern Med. 2019;34(3):458–63.
- Collins M, Long R, Page A, Popay J, Lobban F. Using the Public Involvement Impact Assessment Framework to assess the impact of public involvement in a mental health research context: a reflective case study. Health Expect Int J Public Particip Health Care Health Policy. 2018;21(6):950–63.
- Staniszewska S, Brett J, Simera I, Seers K, Mockford C, Goodlad S, et al. GRIPP2 reporting checklists: tools to improve reporting of patient and public involvement in research. BMJ. 2017;358;j3453.
- Jones J, Cowe M, Marks S, McAllister T, Mendoza A, Ponniah C, et al. Reporting on patient and public involvement (PPI) in research publications: using the GRIPP2 checklists with lay co-researchers. Res Involv Engagem. 2021;7(1):52.
- 14. Cochrane Collaboration. Involving People: A learning resource for systematic review authors. Involving People: a learning resource for systematic review authors. [cited 2023 Oct 15]. Available from: https://training.cochr ane.org/involving-people.
- Greenhalgh T, Hinton L, Finlay T, Macfarlane A, Fahy N, Clyde B, et al. Frameworks for supporting patient and public involvement in research: systematic review and co-design pilot. Health Expect Int J Public Particip Health Care Health Policy. 2019;22(4):785–801.
- Vale CL, Tierney JF, Spera N, Whelan A, Nightingale A, Hanley B. Evaluation of patient involvement in a systematic review and meta-analysis of individual patient data in cervical cancer treatment. Syst Rev. 2012;1(1):23.
- Galante J, Friedrich C, Dawson AF, Modrego-Alarcón M, Gebbing P, Delgado-Suárez I, et al. Mindfulness-based programmes for mental health promotion in adults in nonclinical settings: a systematic review and meta-analysis of randomised controlled trials. PLOS Med. 2021;18(1):e1003481.
- Galante J, Friedrich C, Dalgleish T, Jones PB, White IR. Systematic review and individual participant data meta-analysis of randomized controlled trials assessing mindfulness-based programs for mental health promotion. Nat Ment Health. 2023;1(7):462–76.
- Galante J, Friedrich C, Dalgleish T, White IR, Jones PB. Mindfulness-based programmes for mental health promotion in adults in non-clinical settings: protocol of an individual participant data meta-analysis of randomised controlled trials. BMJ Open. 2022;12(4):e058976.
- 20. Cochrane Training. Understanding evidence-based healthcare [Internet]. [cited 2023 Oct 15]. Available from:https://training.cochrane.org/resou rce/understanding-evidence-based-healthcare-foundation-action.
- Veritas Health Innovation. Covidence [Internet]. Melbourne, Australia: Veritas Health Innovation; [cited 2023 Oct 15]. Available from: https:// www.covidence.org/.

- 22. Brett J, Staniszewska S, Mockford C, Herron-Marx S, Hughes J, Tysall C, et al. A Systematic review of the impact of patient and public involvement on service users, researchers and communities. Patient Patient-Centered Outcomes Res. 2014;7(4):387–95.
- Ocloo J, Garfield S, Franklin BD, Dawson S. Exploring the theory, barriers and enablers for patient and public involvement across health, social care and patient safety: a systematic review of reviews. Health Res Policy Syst. 2021;19(1):8.
- Boden C, Edmonds AM, Porter T, Bath B, Dunn K, Gerrard A, et al. Patient partners' perspectives of meaningful engagement in synthesis reviews: a patient-oriented rapid review. Health Expect. 2021;24(4):1056–71.
- 25. Coon JT, Gwernan-Jones R, Moore D, Richardson M, Shotton C, Pritchard W, et al. End-user involvement in a systematic review of quantitative and qualitative research of non-pharmacological interventions for attention deficit hyperactivity disorder delivered in school settings: reflections on the impacts and challenges. Health Expect. 2016;19(5):1084–97.
- Morley RF, Norman G, Golder S, Griffith P. A systematic scoping review of the evidence for consumer involvement in organisations undertaking systematic reviews: focus on Cochrane. Res Involv Engagem. 2016;2(1):36.
- Hoffmann F, Allers K, Rombey T, Helbach J, Hoffmann A, Mathes T, et al. Nearly 80 systematic reviews were published each day: observational study on trends in epidemiology and reporting over the years 2000–2019. J Clin Epidemiol. 2021;138:1–11.
- Hoddinott P, Pollock A, O'Cathain A, Boyer I, Taylor J, MacDonald C, et al. How to incorporate patient and public perspectives into the design and conduct of research. F1000Res. 2018;7:752.
- Staniszewska S, Denegri S, Matthews R, Minogue V. Reviewing progress in public involvement in NIHR research: developing and implementing a new vision for the future. BMJ Open. 2018;8(7):e017124.
- Amri MM, Abed SA. The data-driven future of healthcare: a review. Mesopotamian J Big Data. 2023;2023:68–74.
- van Dijk SHB, Brusse-Keizer MGJ, Bucsán CC, van der Palen J, Doggen CJM, Lenferink A. Artificial intelligence in systematic reviews: promising when appropriately used. BMJ Open. 2023;13(7):e072254.
- 32. Cipriani A, Seedat S, Milligan L, Salanti G, Macleod M, Hastings J, et al. New living evidence resource of human and non-human studies for early intervention and research prioritisation in anxiety, depression and psychosis. BMJ Ment Health [Internet]. 2023 [cited 2023 Aug 2];26(1). Available from: https://mentalhealth.bmj.com/content/26/1/e300759.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.